Author's response to reviews

Title: Caregivers' active role in palliative home care - to encourage or to dissuade? A qualitative descriptive study

Authors:

Anna Weibull (daugaardweibull@dadlnet.dk)
Frede Olesen (fo@alm.au.dk)
Mette Asbjoern Neergaard (man@alm.au.dk)

Version: 4 Date: 3 September 2008

Author's response to reviews: see over
Concerning MS: 7458005661924524

Caregivers' active role in palliative home care - to encourage or to dissuade?
A qualitative descriptive study
Anna Weibull, Frede Olesen, Mette Asbjoern Neergaard

Dear Editor and Editorial Production Team

We are pleased that our article are now accepted

Below are our changes to the Editorial Production Team’s comments. Our changes are described in red.

On behalf of the authors,
Yours Sincerely

Mette Asbjørn Neergaard
GP, ph.d.-student
Phone: +45 8942 6028
Email: man@alm.au.dk

Formatting changes requested
-----------------------------
Major Revisions
Affiliations - Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution. Can we also ask you to link each author to his or her corresponding affiliation with the use of superscript numbers? Please do not use full capitals for country names, and please do not include full stops in abbreviated country names.
The affiliations has been changed according to the above.

Tables - Tables must be created using a Tables tool. Tables must be divided into rows and columns and be oriented in a portrait page layout. Tables generated with tabbed text are not acceptable. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. Please do not include vertical lines in your tables.
The tables has been changed according to the above.

Minor Revisions
References- Reference XX has not extracted the authors names correctly. Commas need to be inserted between the surname and forename initial. When that's done it will extract correctly, for example Lewis B; Coates R will extract as L B; C R If you change that to: Lewis, B; Coates, R Then it will extract properly.
The references has been changed according to the above.